Author’s response to reviews

Title: One-year Versus Five-year Hospital Readmission after Ischemic Stroke and TIA

Authors:

Anna Therese Bjerkreim (anna.bjerkreim@gmail.com)
Halvor Naess (halvor.ness@helse-bergen.no)
Andrej Netland Khanevski (andrej.netland.khanevski@ihelse.net)
Lars Thomassen (ltho@haukeland.no)
Ulrike Waje-Andreassen (ulrike.waje-andreassen@helse-bergen.no)
Nicola Logallo (nicola.logallo@helse-bergen.no)
Nicola Logallo (nicola.logallo@helse-bergen.no)

Version: 2 Date: 22 Jan 2019

Author’s response to reviews:

We thank you for the opportunity to submit the revised version of our manuscript. We have addressed the issues and suggestions made by the senior assistant editor in the manuscript.

Senior Assistant Editor Comments:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

We have added the email addresses for all authors in corresponding order at the bottom of the title page.

2. Please remove the word count, tables and figures summary from the title page.
We have removed this from the title page

3. Please move the 'Keywords' to after the abstract.

“Keywords” has been moved to after the abstract

4. Thank you for stating that "written informed consent was obtained from all patients or their legally authorized representatives." We kindly ask that you include a statement on the criteria that determined whether consent could be obtained directly from the patient or from a family member or other legal guardian, and confirm that that these criteria were approved by the ethics committee. Please add this information to the Ethics and consent to participate section.

We have included a statement on the criteria for informed consent and confirmed that these were approved by the ethics committee in the methods and under “Ethics and consent to participate”

5. We note that the individual contributions of LT are currently missing from the Authors' Contributions section. Please ensure that all authors are present in this section.

Thank you for noticing us. The contributions of LT (Lars Thomassen) have now been added to this section.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer to Senior Assistant Editor: We have uploaded a single, final, and clean version of the manuscript. We hope that you will consider the manuscript "One-year Versus Five-year Hospital Readmission after Ischemic Stroke and TIA", for publication in BMC Neurology.
Sincerely

(for the authors)

Anna T. Bjerkreim

Research fellow, Neurovascular section